Author's response to reviews

Title: The G-Protein beta 3 subunit 825 TT genotype is associated with epigastric pain syndrome-like dyspepsia

Authors:

Tadayuki Oshima (t-oshima@hyo-med.ac.jp)
Shigemi Nakajima (shigemi@ares.eonet.ne.jp)
Tetsuji Yokoyama (tyokoya@niph.go.jp)
Fumihiko Toyoshima (f-toyo@hyo-med.ac.jp)
Jun Sakurai (jsakurai@hyo-med.ac.jp)
Junji Tanaka (junji-t@hyo-med.ac.jp)
Toshihiko Tomita (tomita@hyo-med.ac.jp)
Yongmin Kim (kimyou@hyo-med.ac.jp)
Kazutoshi Hori (k-hori@hyo-med.ac.jp)
Takayuki Matsumoto (matsut@hyo-med.ac.jp)
Hiroto Miwa (miwahgi@hyo-med.ac.jp)

Version: 3 Date: 25 January 2010

Author's response to reviews: see over
The BMC Series Editorial Production Team
Tel: +44 (0) 20 3192 2014

January 25, 2010

To whom it may concern,

Thank you for considering our manuscript (MS: 4814861213084556) “The G-Protein beta 3 subunit 825 TT genotype is associated with epigastric pain syndrome-like dyspepsia”. We have responded to the reviewers’ comments and have provided a point-by-point response. We hope that you will now find the suitable formatted manuscript for publication in BMC Medical Genetics. Thank you for your time and attention in the handling of our manuscript.

Sincerely,

Hiroto Miwa, M.D., Ph.D.
Division of Upper Gastroenterology
Department of Internal Medicine
Hyogo College of Medicine
1-1 Mukogawa-cho, Nishinomiya, Hyogo 663-8501, Japan
TEL: +81-798-45-6665
FAX: +81-798-45-6661
E-mail: miwahgi@hyo-med.ac.jp
Major revisions (we require the author to make these changes)
Tables: tables should fit onto the width of an A4 portrait page.
We have changed the format.

Minor revisions

Affiliations: Each affiliation should be written in full, in the following format:
Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution
We have corrected them.

Header: Please remove 'Oshima et al' from the top of each page.
We have removed them.

Coloured text: Please present all text in black.
We have corrected them.

Abbreviations: Please include these as a paragraph, not a list.
We have corrected them.